Daily Status 24 October

10/24/13
Summary:

® |OP last night, Dave says it was a good night
® cloudy this morning

Actions past 24 hours:

® SO and SC went to floor to fix P.sw.flr, battery replaced
® SO tried to improve RF comms for P.nne, refer to detail entry
® removed diode from soil battery system at near

To dos:
® SO looking into adding soil sensors to base
Sensor status:

T/RH: ok

P: ok

csat u,v: ok
csat Idiag: ok
csat w, tc: ok
kh2o: ok
motes: ok
Wetness: ok
radiation: ok
Tsoil: ok
Gsoil: ok
Qsoil: ok. Offsets based on yesterday's gravimetric samples make the data look reasonable (for the entire experiment).
Cvsoil: ok.
2D sonic: ok
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